

# GOPEN ACCESS

**Citation:** Barry H, Mutua G, Kibuuka H, Anywaine Z, Sirima SB, Meda N, et al. (2021) Safety and immunogenicity of 2-dose heterologous Ad26. ZEBOV, MVA-BN-Filo Ebola vaccination in healthy and HIV-infected adults: A randomised, placebo-controlled Phase II clinical trial in Africa. PLoS Med 18(10): e1003813. https://doi.org/10.1371/journal.pmed.1003813

Academic Editor: James G. Beeson, Burnet Institute, AUSTRALIA

Received: January 13, 2021

Accepted: September 13, 2021

Published: October 29, 2021

**Copyright:** © 2021 Barry et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Janssen has an agreement with the Yale Open Data Access (YODA) Project to serve as the independent review panel for evaluation of requests for clinical study reports and participant level data from investigators and physicians for scientific research that will advance medical knowledge and public health. Data will be made available following publication and approval by YODA of any formal requests with a defined analysis plan. For more information on this process RESEARCH ARTICLE

Safety and immunogenicity of 2-dose heterologous Ad26.ZEBOV, MVA-BN-Filo Ebola vaccination in healthy and HIV-infected adults: A randomised, placebo-controlled Phase II clinical trial in Africa

Houreratou Barry<sup>®1</sup>, Gaudensia Mutua<sup>2</sup>, Hannah Kibuuka<sup>3</sup>, Zacchaeus Anywaine<sup>9</sup>, Sodiomon B. Sirima<sup>5</sup>, Nicolas Meda<sup>1</sup>, Omu Anzala<sup>2</sup>, Serge Eholie<sup>6</sup>, Christine Bétard<sup>7</sup>, Laura Richert<sup>7,8</sup>, Christine Lacabaratz<sup>8,9</sup>, M. Juliana McElrath<sup>10</sup>, Stephen De Rosa<sup>10</sup>, Kristen W. Cohen<sup>10</sup>, Georgi Shukarev<sup>11</sup>, Cynthia Robinson<sup>11</sup>, Auguste Gaddah<sup>12</sup>, Dirk Heerwegh<sup>12</sup>, Viki Bockstal<sup>11</sup>, Kerstin Luhn<sup>11</sup>, Maarten Leyssen<sup>11</sup>, Macaya Douoguih<sup>11‡</sup>, Rodolphe Thiébaut<sup>7,8‡</sup>, the EBL2002 Study group<sup>11</sup>

 Centre MURAZ, Bobo-Dioulasso, Burkina Faso, 2 KAVI—Institute of Clinical Research University of Nairobi, Nairobi, Kenya, 3 Makerere University—Walter Reed Project, Kampala, Uganda, 4 Medical Research Council/Uganda Virus Research Institute and London School of Hygiene and Tropical Medicine Uganda Research Unit, Entebbe, Uganda, 5 Centre National de Recherche et de Formation sur le Paludisme (CNRFP), Unité de Recherche Clinique de Banfora, Ouagadougou, Burkina Faso, 6 Unit of Infectious and Tropical Diseases, BPV3, Treichville University Teaching Hospital, Abidjan, Côte d'Ivoire, 7 Univ. Bordeaux, Inserm, Bordeaux Population Health Research Center, UMR 1219; Inria SISTM team; CHU Bordeaux; CIC 1401, EUCLID/F-CRIN Clinical Trials Platform, F-33000, Bordeaux, France, 8 Vaccine Research Institute (VRI), Créteil, France, 9 Université Paris-Est Créteil, Faculté de Médecine, INSERM U955, Team 16, Créteil, France, 10 Vaccine and Infectious Disease Division, Fred Hutchinson Cancer Research Center, Seattle, Washington, United States of America, 11 Janssen Vaccines and Prevention, Leiden, the Netherlands, 12 Janssen Research & Development, Beerse, Belgium

‡ These authors are joint senior authors on this work.
¶ Membership of the EBL2002 Study group is provided in the Acknowledgements.

\* rodolphe.thiebaut@u-bordeaux.fr

# Abstract

## Background

We investigated safety, tolerability, and immunogenicity of the heterologous 2-dose Ebola vaccination regimen in healthy and HIV-infected adults with different intervals between Ebola vaccinations.

## Methods and findings

In this randomised, observer-blind, placebo-controlled Phase II trial, 668 healthy 18- to 70year-olds and 142 HIV-infected 18- to 50-year-olds were enrolled from 1 site in Kenya and 2 sites each in Burkina Faso, Cote d'Ivoire, and Uganda. Participants received intramuscular Ad26.ZEBOV followed by MVA-BN-Filo at 28-, 56-, or 84-day intervals, or saline. Females represented 31.4% of the healthy adult cohort in contrast to 69.7% of the HIV-infected cohort. A subset of healthy adults received booster vaccination with Ad26.ZEBOV or saline at Day 365. Following vaccinations, adverse events (AEs) were collected until 42 days post or to make a request, please visit The Yoda Project site at <u>http://yoda.yale.edu</u>. The data sharing policy of Janssen Pharmaceutical Companies of Johnson & Johnson is available at <u>https://www.janssen.</u> com/clinical-trials/transparency.

**Funding:** This work has been supported by Janssen Vaccines and Prevention B.V. and the European Commission IMI2 programme under EBOVAC2 (GA 115861), part of the Innovative Medicines Initiative Ebola+ Program and the National Institute for Health Research (NIHR) Oxford Biomedical Research Centre. Eight authors (GS, CR, AG, DH, VB, KL, ML, and MD) are fulltime employees of the study sponsor (Janssen Vaccines and Prevention) and were involved in study design, data analysis, data interpretation, and writing of the manuscript. The sponsor performed data management and data analyses. Funding for medical writing assistance was provided by Janssen Vaccines and Prevention.

Competing interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: HB, CB, LR reports grants from IMI2-2 [Grant Agreement EBOVAC2 (No.115861) from the Innovative Medicines Initiative 2 Joint Undertaking which receives support from the European Union's Horizon 2020 research and innovation programme], during the conduct of the study. SBS reports grants from Janssen Vaccines and Prevention, during the conduct of the study. GS, CR, AG, DH, VB, KL, ML and MD were full-time employees of Janssen, Pharmaceutical Companies of Johnson & Johnson at the time of the study, and declared ownership of shares in Janssen, Pharmaceutical Companies of Johnson & Johnson. All other authors have nothing to disclose.

Abbreviations: Ad26 VNA, Ad26-specific virus neutralisation assay; AE, adverse event; CI, confidence interval; DRC, Democratic Republic of the Congo; EBOV, Ebola virus; EBOV GP, Ebola virus glycoprotein; ELISA, enzyme-linked immunosorbent assay; EU/mL, ELISA units/mL; ELISpot, enzyme-linked immunospot; FANG, Filovirus Animal Non-Clinical Group; FDA, Food and Drug Administration; GMC, geometric mean concentration; GMT, geometric mean titre; GP, glycoprotein; IC<sub>50</sub>, half maximal inhibitory concentration; ICS, intracellular cytokine staining; IgG, immunoglobulin G; IWRS, interactive web response system; LLOQ, lower limit of quantification; psVNA, pseudovirion neutralisation assay; SAE, serious adverse event; SAGE, Strategic Advisory Group of Experts on Immunization; SD, standard deviation; URTI, upper respiratory tract infection; WHO, World Health Organization.

last vaccination and serious AEs (SAEs) were recorded from signing of the ICF until the end of the study. The primary endpoint was safety, and the secondary endpoint was immunogenicity. Anti-Ebola virus glycoprotein (EBOV GP) binding and neutralising antibodies were measured at baseline and at predefined time points throughout the study.

The first participant was enrolled on 9 November 2015, and the date of last participant's last visit was 12 February 2019. No vaccine-related SAEs and mainly mild-to-moderate AEs were observed among the participants. The most frequent solicited AEs were injection-site pain (local), and fatigue, headache, and myalgia (systemic), respectively. Twenty-one days post-MVA-BN-Filo vaccination, geometric mean concentrations (GMCs) with 95% confidence intervals (CIs) of EBOV GP binding antibodies in healthy adults in 28-, 56-, and 84-day interval groups were 3,085 EU/mL (2,648 to 3,594), 7,518 EU/mL (6,468 to 8,740), and 7,300 EU/mL (5,116 to 10,417), respectively. In HIV-infected adults in 28- and 56-day interval groups, GMCs were 4,207 EU/mL (3,233 to 5,474) and 5,283 EU/mL (4,094 to 6,817), respectively. Antibody responses were observed until Day 365. Ad26.ZEBOV booster vaccination after 1 year induced an anamnestic response.

Study limitations include that some healthy adult participants either did not receive dose 2 or received dose 2 outside of their protocol-defined interval and that the follow-up period was limited to 365 days for most participants.

## Conclusions

Ad26.ZEBOV, MVA-BN-Filo vaccination was well tolerated and immunogenic in healthy and HIV-infected African adults. Increasing the interval between vaccinations from 28 to 56 days improved the magnitude of humoral immune responses. Antibody levels persisted to at least 1 year, and Ad26.ZEBOV booster vaccination demonstrated the presence of vaccination-induced immune memory. These data supported the approval by the European Union for prophylaxis against EBOV disease in adults and children  $\geq$ 1 year of age.

## **Trial registration**

ClinicalTrials.gov NCT02564523

## Author summary

## Why was this study done?

- With Ebola outbreaks increasing, there is an unmet medical need for a prophylactic vaccine to prevent and mitigate Ebola outbreaks.
- To address the urgent medical need during the 2014 to 2016 outbreak, the clinical development of the 2-dose vaccine regimen comprising of Ad26.ZEBOV and MVA-BN-Filo was accelerated. This Phase II study was part of this accelerated program, evaluating the safety and immunogenicity of the 2-dose vaccine regimen in healthy and HIV-infected African adults, with 28-, 56-, and 84-day intervals between doses.

• The study was amended to evaluate safety and immunogenicity of a booster vaccination with Ad26.ZEBOV, administered approximately 1 year after the first vaccination, in healthy adults.

## What did the researchers do and find?

- We conducted a randomised trial to assess the safety and the immunogenicity of the Ad26.ZEBOV, MVA-BN-Filo vaccine regimen in 3 different vaccination intervals in healthy and HIV-infected adults.
- The vaccine regimen was well tolerated and induced marked immune responses; the highest humoral responses were observed after vaccination with 56-day and 84-day intervals.
- Anamnestic responses were observed in all healthy participants receiving Ad26.ZEBOV as booster at 1 year after the first dose.

## What do these findings mean?

- Our results demonstrate that the Ad26.ZEBOV, MVA-BN-Filo vaccine regimen is safe and immunogenic in healthy and HIV-infected adults and induces immune memory that can rapidly be reactivated.
- Our findings support the prophylactic use of the 2-dose Ad26.ZEBOV, MVA-BN-Filo vaccine regimen against Ebola infection in African adult populations.
- The 2-dose vaccine regimen comprising of Ad26.ZEBOV and MVA-BN-Filo has received marketing authorisation under exceptional circumstances for prophylactic use against EVD in adults and children ≥1 year old within the European Union.

## Introduction

Since its first identification as the cause of a haemorrhagic fever in 1976, Ebola virus (EBOV) has become increasingly recognised as a widely dispersed pathogen across the African continent [1], with an increasing number of outbreaks. In most cases, infection starts with nonspecific influenza-like symptoms—fatigue, myalgia, arthralgia, headache, and fever. These may be followed by more severe clinical manifestations ultimately leading to multi-organ dysfunction, vascular leakage, and intense generalised bleeding resulting in death within 5 to 8 days of symptom onset [2]. In the biggest outbreak to date in Guinea, Liberia and Sierra Leone in 2014 to 2016, there were 28,616 cases and 11,310 deaths [3]. The 10th outbreak in Democratic Republic of the Congo (DRC) from 2018 to 2020 in the Kivu area was the second largest in history with 3,470 cases, including 2,287 deaths [4,5]. Several EBOV vaccines are in development [6] and target the viral surface glycoprotein (GP). The main difference between vaccines is the vector used to deliver the antigen. In a ring vaccination strategy, efficacy is assessed through the vaccination of contacts, and contacts of contacts, of individuals with confirmed cases of EBOV disease. This strategy was applied in the 2014 to 2016 West African outbreak in Guinea and the 2018 to 2020 DRC outbreak, in which 1 dose of a recombinant replication-competent

vesicular stomatitis viral vectored vaccine (rVSV-ZEBOV-GP) expressing the GP of the Kikwit variant of the Zaire species demonstrated 100% [7] and 97.5% [8] efficacy from 10 days after vaccination, respectively. This vaccine (Ervebo) has been approved by the Food and Drug Administration (FDA) [9] for use in adults  $\geq$ 18 years old and received conditional approval by the European Medicines Agency [10]. It was prequalified by the World Health Organization (WHO) [11] and is recommended by the WHO Strategic Advisory Group of Experts on Immunization (SAGE) for high-risk populations in response to the 2018 to 2020 outbreak in the DRC [12].

Recently, the heterologous 2-dose regimen of Ad26.ZEBOV (Zabdeno) and MVA-BN-Filo (Mvabea) received approval by the European Commission under exceptional circumstances for use in children and adults [13]. This regimen is recommended by WHO SAGE for prophylaxis and evaluation in lower-risk populations [12]. Following WHO SAGE recommendation, the heterologous regimen is being studied in the DRC and in an ongoing vaccination campaign in Rwanda in response to the 2018 to 2020 outbreak [14].

In Phase I studies, this regimen had an acceptable safety profile and was immunogenic in healthy 18- to 50-year-old volunteers in England [15], Kenya [16], and Tanzania and Uganda [17], inducing both humoral and cellular immune responses against EBOV GP. These data encouraged further evaluation in Phase II and III studies.

We present a Phase II study to determine the safety, tolerability, and immunogenicity of this heterologous 2-dose regimen, with various intervals between the 2 doses in the target population, i.e., healthy African adults, adolescents, and children. We also included a cohort of HIV-infected adults to ensure that the vaccine regimen is safe and immunogenic in this population. Finally, we explored the effect of an Ad26.ZEBOV booster vaccination administered 1 year post-dose 1 in a subset of healthy adults. This paper presents the results from the healthy and HIV-infected adult cohorts. Data from the adolescent and paediatric populations will be published separately.

## Methods

#### Study overview

This Phase II randomised, observer-blind, placebo-controlled study was conducted in 7 sites in Africa [Burkina Faso (Bobo-Dioulasso, Banfora); Côte d'Ivoire (Abidjan, Toupah/ Ousrou); Kenya (Nairobi); Uganda (Masaka, Kampala)] between November 2015 and February 2019. The protocol was approved by local and national independent Ethics Committees and Institutional Review Boards, and the study was done according to the Declaration of Helsinki and International Conference on Harmonisation Good Clinical Practice Guide-lines. All adult participants supplied written informed consent before enrolment. An independent data monitoring committee was established to assess the safety data regularly during the study. The study was registered at <u>ClinicalTrials.gov</u> NCT02564523. The study protocol, including the CONSORT checklist, can be found in <u>S1 Study Protocol</u> and <u>S1</u> CONSORT Checklist.

The primary objective was to assess the safety and tolerability of 3 vaccination schedules with Ad26.ZEBOV administered on Day 1 and MVA-BN-Filo on Days 29, 57, or 85 in healthy African adults, and of 2 schedules with Ad26.ZEBOV on Day 1 and MVA-BN-Filo on Days 29 or 57 in HIV-infected African adults. Secondary objectives included immunogenicity assessments such as EBOV GP–specific immunoglobulin G (IgG) binding antibodies 21 days post-dose 2 in healthy and HIV-infected adults, and the safety and tolerability of an Ad26.ZEBOV booster dose on Day 365 in a subset of 90 healthy adults. Exploratory objectives included assessments of binding antibodies at other time points in healthy (including after the booster

dose) and HIV-infected adults and measurement of EBOV GP-specific neutralising antibodies and cellular immune responses in subsets of healthy adults.

#### Study participants, randomisation, and blinding

Study participants were recruited from the general population and from HIV–positive cohorts in Africa. Information regarding the study was shared through community meetings, posters, and public conferences where volunteers were invited to study sites. After screening by the investigators at study sites, eligible healthy adults aged 18 to 70 years were randomised to 3 groups (1:1:1) in parallel using randomly permuted blocks. Groups 1, 2, and 3 correspond to 28-, 56-, and 84-day intervals, respectively. For the healthy adult cohort, there was stratification by age (18 to 50 years and >50 years). The interactive web response system (IWRS; prepared by Signant Health) assigned a unique code, which dictated the assignment and matching vaccination schedule for a participant. These codes were maintained within the IWRS and were not provided to the investigators. During the study pause (detailed in the results section), investigators were instructed not to proceed with any randomisation until the pause was lifted.

Healthy adults were enrolled in parallel to Groups 1 to 3 until a target of 132 participants had been randomised to Group 3. Further enrolment continued to Groups 1 and 2 with 1:1 randomisation until each group contained 264 participants. When 7-day safety data were available from 25% of healthy adults, enrolment of the HIV-infected cohort started, consisting of 142 HIV-infected adults aged 18 to 50 years randomised 1:1 to 2 groups to receive the 2 vaccinations with a 28- or 56-day interval. Participants in each of the 5 groups in healthy and HIV-infected cohorts were further randomised 5:1 to receive active vaccines or placebo in the different regimens. The first 90 healthy adult participants in Groups 1 and 2 (vaccinees, n = 73or placebo, n = 17) who provided consent received an Ad26.ZEBOV booster dose or placebo on Day 365. Participants, investigators, and study staff remained blinded to the allocation of investigational products throughout the study. Vaccines and placebo were prepared by a site pharmacist who was the only unblinded member of staff. The pharmacist received the randomisation number and allocated the right study vaccine to the participant. Masking tape was used to cover the dispensing syringes containing the study vaccine/placebo allocated to each study participant.

Eligible volunteers were confirmed healthy based on physical examination, ECG, vital signs, and clinical laboratory assessments performed at screening and their medical history. HIV-infected volunteers had to have been diagnosed at least 6 months previously, have a CD4+ cell count >350 cells/ $\mu$ L, be in an otherwise reasonably good medical condition without an AIDS-defining diagnosis or a clinically significant disease, and be on a stable regimen of antiretroviral therapy. Full details of inclusion and exclusion criteria are supplied in S1 Text.

#### Sample size determination

The overall planned sample size for healthy and HIV-infected cohorts included 792 participants who were to receive either the 2-dose heterologous vaccination regimen of Ad26. ZEBOV and MVA-BN-Filo or placebo to substantially contribute to the overall safety database of the regimen. In the healthy adult cohort, a total of 550 participants were to receive Ad26. ZEBOV and MVA-BN-Filo across the different vaccination schedules (220 participants in 28-day interval and 220 participants in 56-day interval; 110 in 84-day interval); 110 participants were to receive placebo. The preplanned smaller sample size target for Group 3 (84-day interval) was selected to provide preliminary information on the effect of a longer interval between doses. The larger sample size target for Groups 1 and 2 was reflective of the schedules for which an indication was being sought. As the study was not powered to test statistical difference between the groups, we consider that these differences in recruitment are unlikely to impact the conclusions of the study. In the HIV-infected adult cohort, a total of 110 participants were to receive Ad26.ZEBOV and MVA-BN-Filo across the different vaccination schedules (55 participants in 28-day interval and 55 in 56-day interval); 22 participants were to receive placebo. Sample size determination was not based on formal statistical hypothesis testing. However, in case a specific adverse event (AE) was not observed, the one-sided 97.5% upper confidence limit of the true incidence rate of this AE was less than 6.5%, 3.3%, 1.7%, and 0.7% for a sample size of 55, 110, 220, and 550 participants, respectively.

#### Vaccines

Ad26.ZEBOV is a monovalent, recombinant, replication-incompetent adenovirus 26-vectored vaccine encoding EBOV Mayinga variant GP. MVA-BN-Filo is a multivalent, recombinant, nonreplicating, modified vaccinia Ankara-vectored vaccine encoding EBOV Mayinga, Sudan virus Gulu, and Marburg virus Musoke variant GPs, as well as Taï Forest virus nucleoprotein [15–17].

Ad26.ZEBOV vaccination with  $5 \times 10^{10}$  viral particles on Day 1 was followed by  $1 \times 10^{8}$  infectious units of MVA-BN-Filo on Day 29 (Group 1), Day 57 (Group 2), or Day 85 (Group 3). The booster vaccination (Day 365) consisted of Ad26.ZEBOV  $5 \times 10^{10}$  viral particles or placebo. Both vaccines and placebo (0.9% saline) were administered by intramuscular injection (0.5 mL) in the deltoid. The second injection was administered in the opposite arm to the first injection.

#### Safety and tolerability assessments

All participants were assessed at 30 and 60 minutes after each vaccination for any immediate AEs. Participants were supplied with diary cards, rulers, and thermometers to measure and record solicited local and systemic AEs and daily body temperature each evening for 7 days. AEs were graded as 1 (mild), 2 (moderate), or 3 (severe) according to the adapted Division of AIDS (DAIDS) Table (2017) [18]. Laboratory assessments of haematology and serum chemistry were done at screening, prevaccination, and 7 days after vaccination and graded according to the FDA toxicity grading scale for healthy adult and adolescent volunteers enrolled in preventive vaccine clinical trials [19]. Any unsolicited AEs, until 42 days after the second vaccination or 28 days after the booster dose, were documented by the investigator. Any serious AEs (SAEs) were recorded throughout the study, with follow-up to 1 year after the last vaccination or event resolution as appropriate.

#### Immunogenicity assessments

Serum samples obtained from all participants at 4 time points were used to assess humoral immune responses as binding IgG antibodies specific to EBOV GP: before the first vaccination on Day 1 and the second vaccination on Days 29, 57 or 85, and 21 days post-dose 2, with a final sample on Day 365 to assess persistence of antibodies. In the booster subset, additional blood samples were obtained 4, 7, 21, and 365 days after receiving the booster dose. EBOV GP–specific binding antibodies were measured at all time points using an EBOV GP Filovirus Animal Non-Clinical Group (FANG) enzyme-linked immunosorbent assay (ELISA) [20] performed at Q<sup>2</sup> Solutions (San Juan Capistrano, CA, USA).

Virus neutralising antibodies were measured in a subset (19%; n = 100/536 [n = 50/268 assigned to the 28-day interval regimen; n = 50/268 assigned to the 56-day interval regimen]) of healthy adults on Day 1, 21 days after MVA-BN-Filo and at 1 year using the EBOV GP

pseudovirion neutralisation assay (psVNA) at Monogram (San Francisco, CA, USA) (**S2 and S3 Texts**). Levels of neutralising antibodies to the Ad26 vector were measured using an Ad26-specific virus neutralisation assay (Ad26 VNA) in a subset (19%; n = 100/536 [n = 50/268 assigned to the 28-day interval regimen; n = 50/268 assigned to the 56-day interval regimen]) of healthy adults and in all HIV-infected participants before vaccination on Day 1 and again on Day 365 (pre-booster) for healthy adults (Janssen Vaccines and Prevention, Leiden, the Netherlands). Intracellular cytokine staining (ICS) and interferon gamma (IFN $\gamma$ ) enzymelinked immunospot (ELISpot) assays were done in a subset of healthy (16%; n = 84/536[n = 46/268 assigned to the 28-day interval regimen; n = 38/268 assigned to the 56-day interval regimen]) and HIV-infected (21%; n = 15/71 assigned to the 56-day interval regimen) participants (see **S2 Text** for selection criteria) on Day 1, 21 days post-dose 2 and 1 year post-dose 1 (HIV Vaccine Trials Network, Seattle, WA, USA), as previously described [15–17,21].

#### Statistical analyses

The study was originally designed as a prospective study with no formal hypothesis testing. Safety analyses were descriptive, based on the full analysis set consisting of all participants who received at least 1 dose of active vaccine or placebo regardless of protocol deviations. The main immunogenicity analysis is based on the per-protocol analysis set. It included all randomised and vaccinated participants who received both vaccinations in the protocol-defined windows, had at least 1 evaluable immunogenicity sample post vaccination, and no major protocol deviations influencing the immune response. A sensitivity analysis was conducted for those who received their second vaccination outside the protocol-defined windows. These participants were part of the immunogenicity analysis set that included all randomised and vaccinated participants who had at least 1 postvaccination (i.e., after date of vaccination) evaluable immunogenicity sample. Binding antibody results were expressed as group geometric mean concentrations (GMCs) of ELISA units with 95% confidence intervals (CIs) at each time point and responder rates, i.e., the percentage of each group with post-vaccination concentrations >2.5-fold the lower limit of quantification [LLOQ; 36.11 ELISA units/mL (EU/mL)] in baseline seronegative individuals or >2.5-fold the baseline value in initially seropositive participants. All values below the LLOQ were imputed with half the LLOQ value. Exploratory analysis responder definitions are reported in <u>S4 Text</u>. A Spearman correlation coefficient was calculated for EBOV GP-specific binding antibody concentrations (FANG ELISA) and psVNA titres 21 days post-dose 2.

While no formal statistical testing was originally planned, statistical analyses were performed post hoc for primary outcomes. The statistical significance was set at *p*-value < 0.05. No test multiplicity adjustments were generally performed.

## Results

#### Participant enrolment and clinical hold

Enrolment of participants began on 9 November 2015 and was paused on 28 April 2016 to permit a thorough investigation of 2 SAEs that occurred in a concomitant study in Europe [22]. At the time of the pause, more than 200 healthy participants had been enrolled, some awaiting their first vaccination, and some already having received the first vaccine. When the independent data monitoring committee and the regulatory authorities agreed to continuation of the study (clinical hold lifted by US FDA on 16 June 2016), participants who had already received 1 dose before the clinical hold were offered the second vaccination out-of-window but were excluded from the per-protocol analysis set. Those who had been enrolled but had not received their first vaccination were rescreened and vaccinated according to their preassigned study schedule. Vaccination of the HIV-infected participants only started after lifting the clinical hold. Monitoring was continued for all participants up to the 365-day follow-up, irrespective of their inclusion in the per-protocol analysis set. The date of last participant last visit was 12 February 2019.

#### **Baseline demographics**

The full analysis set included 668 healthy adults (559 vaccinees, 109 placebo) and 142 HIV-infected adults (118 vaccinees, 24 placebo), who received at least 1 dose of Ad26. ZEBOV or placebo (Fig 1). Participant demographic data are shown in Table 1. Among these, 400 healthy adults and 140 HIV-infected adults were included in the per-protocol analysis set (Table 1).

#### Safety

Safety analyses are based on the full analysis set. There were 27 SAEs (which occurred in 24 participants) including 1 death overall during the study (**Table A in <u>S1 Data</u>**). One HIV-infected participant died due to alcohol poisoning more than 4 weeks after receipt of MVA-BN-Filo. No SAE was considered to be vaccine-related by the investigators.

Both vaccines were well tolerated in healthy and HIV-infected adults (Table 2). Reactogenicity results were similar when healthy adults were stratified to 18 to 50 years and >50 years age groups (Table B in S1 Data).

**Safety of 2-dose vaccination primary regimen.** Overall, in healthy adults, solicited local AEs were reported in 307/559 (54.9%), 296/517 (57.2%), 47/109 (43.1%), and 34/99 (34.3%) participants following Ad26.ZEBOV, MVA-BN-Filo, first placebo, and second placebo injections, respectively. The most common solicited local AE was mild or moderate injection-site pain. In HIV-infected adults, solicited local AEs were reported in 69/118 (58.5%), 51/117 (43.6%), 6/24 (25.0%), and 4/24 (16.7%) participants following Ad26.ZEBOV, MVA-BN-Filo, first placebo, and second placebo injections, respectively. The event rate following Ad26. ZEBOV in healthy adults (54.9%) was not statistically different (*p*-value = 0.54) from that of the HIV-infected adults (58.5%; **Table C in S1 Data**); following MVA-BN-Filo, a significant difference (57.2% versus 43.6%, respectively, *p*-value = 0.01) was observed. In the healthy



Fig 1. Study disposition of healthy adults and HIV-infected adults. n = number of participants.

https://doi.org/10.1371/journal.pmed.1003813.g001

Characteristic			Healthy	y adults			HIV-infected adults					
	Group 1: 28-	day interval	Group 2: 56-	day interval	Group 3: 84-	day interval	Group 1: 28-	day interval	Group 2: 56-day interval			
	Active vaccines	Placebo	Active vaccines	Placebo	Active vaccines	Placebo	Active vaccines	Placebo	Active vaccines	Placebo		
N =	225	43	224	44	110	22	59	12	59	12		
Sex, n (%)												
Male	152 (67.6)	28 (65.1)	153 (68.3)	28 (63.6)	83 (75.5)	14 (63.6)	20 (33.9)	4 (33.3)	17 (28.8)	2 (16.7)		
Female	73 (32.4)	15 (34.9)	71 (31.7)	16 (36.4)	27 (24.5)	8 (36.4)	39 (66.1)	8 (66.7)	42 (71.2)	10 (83.3)		
Ethnicity <sup>‡</sup> , n (%)												
Black	223 (99.1)	42 (97.7)	224 (100)	43 (97.7)	109 (99.1)	22 (100)	59 (100)	12 (100)	59 (100)	12 (100)		
Asian	2 (0.9)	1 (2.3)	0	0	1 (0.9)	0	0	0	0	0		
White	0	0	0	1 (2.3)	0	0	0	0	0	0		
Age (years)												
Mean (SD)	33.3 (12.4)	32.0 (10.4)	33.3 (11.5)	33.3 (11.6)	31.4 (11.5)	33.7 (12.3)	38.8 (6.6)	34.3 (7.7)	39.0 (6.7)	42.1 (5.0)		
Median (range)	29.0 (18-69)	30.0 (19-60)	30.0 (18-66)	30.0 (18-65)	27.0 (18-67)	31.0 (19–66)	40.0 (21-49)	36.5 (22-48)	39.0 (18-49)	41.5 (34-50)		
Body mass index (k	(g/m <sup>2</sup> )											
Mean (SD)	23.0 (3.95)	23.4 (4.55)	23.4 (3.65)	22.9 (4.18)	23.0 (4.29)	25.1 (6.10)	24.5 (4.51)	23.0 (3.88)	24.0 (4.51)	24.5 (3.78)		
Per-protocol analys	sis set <sup>#</sup>											
N# =	173	32	137	24	27	7	58	11	59	12		

#### Table 1. Baseline characteristics of the participants in the full analysis set.\*

\*Participants who received at least 1 dose of the study vaccines or placebo.

<sup>\*</sup>Ethnicity was recorded as race within the database.

<sup>#</sup>The numbers of participants who received the study vaccines or placebo within the predefined time windows and supplied at least 1 postvaccination sample with no protocol deviations.

N, number of participants who received at least 1 dose of the study vaccines or placebo; SD, standard deviation.

https://doi.org/10.1371/journal.pmed.1003813.t001

adults, 2/559 (0.4%) and 4/517 (0.8%) participants reported Grade 3 local AEs following Ad26. ZEBOV and MVA-BN-Filo injections, respectively; no Grade 3 local AEs were reported by placebo recipients or HIV-infected adults.

In healthy adults, solicited systemic AEs were reported in 361/559 (64.6%), 306/517 (59.2%), 67/109 (61.5%), and 49/99 (49.5%) participants following Ad26.ZEBOV, MVA-BN-Filo, first placebo, and second placebo injections, respectively. The most frequent systemic AEs were fatigue, headache, and myalgia. Compared to the healthy adults, there was no significant difference (*p*-value = 0.53 following Ad26.ZEBOV; *p*-value = 0.06 following MVA-BN-Filo; **Table C in S1 Data**) in reactogenicity profile in HIV-infected adults in whom systemic AEs were reported in 80/118 (67.8%), 58/117 (49.6%), 9/24 (37.5%), and 10/24 (41.7%) participants following Ad26.ZEBOV, MVA-BN-Filo, first placebo, and second placebo injections, respectively. Grade 3 systemic AEs were reported in 0% to 4.2% of healthy and HIV-infected adults following Ad26.ZEBOV, MVA-BN-Filo, or placebo injections.

In healthy adults, fever ( $\geq$ 38.0°C) occurred in 25/559 (4.5%), 33/517 (6.4%), 4/109 (3.6%), and 3/99 (3.0%) participants following Ad26.ZEBOV, MVA-BN-Filo, first placebo, and second placebo injections, respectively. Grade 3 fever (>38.9°C) was reported in 4/559 (0.7%), 9/517 (1.7%), 0/109 (0.0%), and 1/99 (1.0%) participants following Ad26.ZEBOV, MVA-BN-Filo, first placebo, and second placebo injections, respectively (**Table D in S1 Data**). In HIV-infected participants, fever was reported in 13/118 (11.0%), 3/117 (2.6%), 2/24 (8.3%), and 3/24 (12.5%) participants following Ad26.ZEBOV, MVA-BN-Filo, first placebo, and second placebo injections, respectively. The fever rate following Ad26.ZEBOV in healthy adults (4.5%) was statistically lower (*p*-value = 0.01) than that of the HIV-infected adults (11.0%; **Table C in S1 Data**);

					Health	y adults	HIV-infected adults								
		Group 1: 28-day interval			2: 56-day erval	Group 3: 84-day interval		All groups		Group 1: 28-day interval		Group 2: 56-day interval		All groups	
		Ad26	Placebo	Ad26	Placebo	Ad26	Placebo	Ad26	Placebo	Ad26	Placebo	Ad26	Placebo	Ad26	Placebo
Post-dose 1	N	225	43	224	44	110	22	559	109	59	12	59	12	118	24
Solicited AE,		161	30	161	31	83	15	405	76	48	7	45	3	93	10
n (%)		(71.6)	(69.8)	(71.9)	(70.5)	(75.5)	(68.2)	(72.4)	(69.7)	(81.4)	(58.3)	(76.3)	(25.0)	(78.8)	(41.7)
Grade 3 <sup>‡</sup>		4 (1.8)	0	6 (2.7)	1 (2.3)	5 (4.5)	1 (4.5)	15 (2.7)	2 (1.8)	1 (1.7)	0	2 (3.4)	0	3 (2.5)	0
Solicited local AE, n (%)		123 (54.7)	16 (37.2)	121 (54.0)	20 (45.5)	63 (57.3)	11 (50.0)	307 (54.9)	47 (43.1)	34 (57.6)	4 (33.3)	35 (59.3)	2 (16.7)	69 (58.5)	6 (25.0)
Grade 3 <sup>‡</sup>		0	0	1 (0.4)	0	1 (0.9)	0	2 (0.4)	0	0	0	0	0	0	0
Solicited systemic AE, n	ı (%)	146 (64.9)	27 (62.8)	140 (62.5)	26 (59.1)	75 (68.2)	14 (63.6)	361 (64.6)	67 (61.5)	44 (74.6)	7 (58.3)	36 (61.0)	2 (16.7)	80 (67.8)	9 (37.5)
Grade 3 <sup>‡</sup>		4 (1.8)	0	6 (2.7)	1 (2.3)	5 (4.5)	1 (4.5)	15 (2.7)	2 (1.8)	1 (1.7)	0	2 (3.4)	0	3 (2.5)	0
Any unsolicite AE, n (%)	d	96 (42.7)	17 (39.5)	68 (30.4)	17 (38.6)	37 (33.6)	6 (27.3)	201 (36.0)	40 (36.7)	25 (42.4)	7 (58.3)	25 (42.4)	3 (25.0)	50 (42.4)	10 (41.7)
Grade 3 <sup>‡</sup>			5	7 (3.1)	4	3 (2.7)	2 (9.1)	26	11	5 (8.5)	2	4 (6.8)	1	9 (7.6)	3
		(7.1) MVA	(11.6) Placebo	MVA	(9.1) Placebo	MVA	Placebo	(4.7) MVA	(10.1) Placebo	(8.3) MVA	(16.7) Placebo	MVA	(8.3) Placebo	(7.0) MVA	(12.5) Placebo
Post-dose 2	N	219	39	200	39	98	21	517	99	58	12	59	12	117	24
Solicited AE,	11	152	15	148	25	73	15	373	55	34	5	31	6	65	11
n (%)		(69.4)	(38.5)	(74.0)	(64.1)	(74.5)	(71.4)	(72.1)	(55.6)	(58.6)	(41.7)	(52.5)	(50.0)	(55.6)	(45.8)
Grade 3 <sup>‡</sup>		5 (2.3)	2 (5.1)	2 (1.0)	1 (2.6)	6 (6.1)	0	13 (2.5)	3 (3.0)	0	0	0	1 (8.3)	0	1 (4.2)
Solicited local n (%)	AE,	126 (57.5)	9 (23.1)	113 (56.5)	16 (41.0)	57 (58.2)	9 (42.9)	296 (57.2)	34 (34.3)	26 (44.8)	2 (16.7)	25 (42.4)	2 (16.7)	51 (43.6)	4 (16.7)
Grade 3 <sup>‡</sup>		2 (0.9)	0	0	0	2 (2.0)	0	4 (0.8)	0	0	0	0	0	0	0
Solicited	(0/)	121	14	121	21	64	14	306	49	32	4	26	6	58	10 (41.7)
systemic AE, n	ı (%)	(55.3)	(35.9)	(60.5)	(53.8)	(65.3)	(66.7)	(59.2)	(49.5)	(55.2)	(33.3)	(44.1)	(50.0)	(49.6)	(41.7)
Grade 3 <sup>‡</sup>		4 (1.8)	2 (5.1)	2 (1.0)	1 (2.6)	5 (5.1)	0	11 (2.1)	3 (3.0)	0	0	0	1 (8.3)	0	1 (4.2)
Any unsolicite	d	74	15	58	13	34	7	166	35	22	4	22	4	44	8
AE, n (%) Grade 3 <sup>‡</sup>		(33.8)	(38.5)	(29.0) 7	(33.3)	(34.7)	(33.3)	(32.1) 15	(35.4)	(37.9)	(33.3) 0	(37.3)	(33.3)	(37.6)	(33.3)
		(2.3)	(2.6)	(3.5)	(5.1)	(3.1)	(4.8)	(2.9)	(4.0)	(1.7)		(5.1)	(8.3)	(3.4)	(4.2)
		Ad26	Placebo	Ad26	Placebo			Ad26	Placebo						
Post-booster	N	34	8	39	9	NA	NA	73	17						
Solicited AE, n (%)		20 (58.8)	3 (37.5)	22 (56.4)	4 (44.4)			42 (57.5)	7 (41.2)						
Grade 3 <sup>‡</sup>		0	0	1 (2.6)	0			1 (1.4)	0						
Solicited local AE, n (%)		18 (52.9)	1 (12.5)	16 (41.0)	3 (33.3)			34 (46.6)	4 (23.5)						
Grade 3 <sup>‡</sup>		0	0	0	0			0	0						
Solicited systemic AE, n (%)		17 (50.0)	2 (25.0)	18 (46.2)	4 (44.4)			35 (47.9)	6 (35.3)						
Grade 3 <sup>‡</sup>		0	0	1 (2.6)	0			1 (1.4)	0						
Any unsolicited AE, n (%)		9 (26.5)	3 (37.5)	14 (35.9)	0			23 (31.5)	3 (17.6)						

#### Table 2. Solicited and unsolicited AEs in the full analysis set after each vaccination dose, no. (%).\*

(Continued)

#### Table 2. (Continued)

	Healthy adults										HIV-infected adults					
	1	Group 1: 28-day interval		Group 2: 56-day interval		Group 3: 84-day interval		All groups		Group 1: 28-day interval		Group 2: 56-day interval		All groups		
Grade 3 <sup>‡</sup>	0	0	3 (7.7)	0			3 (4.1)	0								

\*Solicited local and systemic AEs were recorded on diary cards by participants for 7 days and unsolicited AEs until Day 28 after each vaccination. Solicited local AEs included pain, erythema, swelling, and pruritis; systemic AEs included nausea/vomiting, fatigue/malaise, headache, myalgia, arthralgia, and chills.

<sup>‡</sup>Grade 3 was defined as symptoms causing inability to perform usual social and functional activities.

Vaccines: Ad26 = Ad26.ZEBOV at a dose of  $5 \times 10^{10}$  vp; MVA = MVA-BN-Filo at a dose of  $1 \times 10^{8}$  Inf.U.

AE, adverse event; N, number of participants with data at that time point; n (%), number (percentage) of participants with one or more events; NA, not applicable.

https://doi.org/10.1371/journal.pmed.1003813.t002

following MVA-BN-Filo, no statistically significant difference (6.4% versus 2.6%, respectively, *p*-value = 0.12) was observed.

Grade 3 fever was reported in 3/118 (2.5%) participants following Ad26.ZEBOV injections, 1/24 (4.2%) participants following second placebo injections and in no participants following MVA-BN-Filo or first placebo injections.

In healthy adults, unsolicited AEs were reported at similar rates in each group; reported in 201/559 (36.0%), 166/517 (32.1%), 40/109 (36.7%), and 35/99 (35.4%) participants following Ad26.ZEBOV, MVA-BN-Filo, first placebo, and second placebo injections, respectively. The most frequent unsolicited AEs were malaria and upper respiratory tract infections (URTIs). The rates of unsolicited AEs reported in HIV-infected adults were 50/118 (42.4%), 44/117 (37.6%), 10/24 (41.7%), and 8/24 (33.3%) in participants following Ad26.ZEBOV, MVA-BN-Filo, first placebo, and second placebo injections, respectively. Compared to the healthy adults, the observed rates in HIV-infected adults were not significantly different following either vaccine (*p*-value = 0.21 following Ad26.EBOV; *p*-value = 0.28 following MVA-BN-filo; **Table C in S1 Data**). The most frequent unsolicited AEs in the HIV-infected adults were URTI and cases of neutropaenia, which were reported equally in vaccine and placebo arms.

**Safety of booster dose.** There was no apparent exacerbation of reactogenicity after the booster dose of Ad26.ZEBOV in healthy adults. Of these 73 participants, 34 (46.6%) and 35 (47.9%) reported local or systemic AEs, respectively, of which only one was Grade 3 (fever).

#### Immunogenicity

Overall, immunogenicity results were similar when healthy adults were stratified to 18 to 50 years and >50 years age groups (Table E in <u>S1 Data</u>).

**Binding antibody responses against EBOV GP.** The immunogenicity analyses were based on the participants in the per-protocol analysis set. Binding antibodies in all placebo groups were very low or undetectable at all time points.

After Ad26.ZEBOV vaccination of healthy adults, there were marked increases in EBOV GP–specific binding antibodies (**Fig 2A**). Prior to dose 2 vaccination, levels at Days 29, 57, and 85 in Groups 1, 2, and 3, respectively, were similar with GMCs of 332 EU/mL (95% CI 282 to 390), 361 EU/mL (95% CI 307 to 423), and 242 EU/mL (95% CI 181 to 323) (**Fig 2A**). The proportion of responders was 77%, 80%, and 81%, respectively (**Table F in S1 Data**). MVA-BN-Filo vaccination elicited further increases in antibody concentrations, reaching GMCs of 3,085 EU/mL (95% CI 2,648 to 3,594), 7,518 EU/mL (95% CI 6,468 to 8,740), and 7,300 EU/mL (95% CI 5,116 to 10,417) in the 28-, 56-, and 84-day interval groups, respectively, 21 days post-dose 2 (**Fig 2A**). The GMCs of binding antibodies were statistically different for

Α



в



**Fig 2. Anti-EBOV GP binding antibody responses. (A)** Healthy adults administered Ad26.ZEBOV on Day 1 and MVA-BN-Filo 28, 56, or 84 days later as indicated. Subsets of 28- and 56-day groups received Ad26.ZEBOV booster on Day 365. "Pre-booster" indicates the pre-booster vaccination measurement observed in participants who received the booster vaccination. **(B)** HIV-infected adults administered with Ad26.ZEBOV on Day 1 and MVA-BN-Filo 28 or 56 days later, as indicated. Responses are expressed as GMCs (ELISA units/mL, 95% CI). Responses in placebo groups are shown as open symbols. Black dotted line represents the LLOQ. The points (symbols) denote GMCs, and error bars denote 95% CIs. Vaccines: Ad26 = Ad26.ZEBOV at a dose of  $5 \times 10^{10}$  vp; MVA = MVA-BN-Filo at a dose of  $1 \times 10^8$  Inf.U. CI, confidence interval; GMC, geometric mean concentration; LLOQ, lower limit of quantification.

https://doi.org/10.1371/journal.pmed.1003813.g002

the 28- and 56-day interval groups (GMC ratio [95% CI] = 0.4 [0.3 to 0.5], *p*-value < 0.001) and the 28- and 84-day interval groups (GMC ratio [95% CI] = 0.4 [0.3 to 0.7], *p*-value < 0.001) but not for 56- and 84-day interval groups (GMC ratio [95% CI] = 1.0 [0.6 to 1.6], *p*-value = 1.00; **Table G in S1 Data**). At this time, responder rates were 98%, 99%, and 100%, respectively (**Table F in S1 Data**). At Day 365, binding antibodies persisted with GMCs ranging from 313 to 363 EU/mL in 80%, 78%, and 88% of participants in the 28-, 56-, and 84-day groups, respectively (**Fig 2A and Table F in S1 Data**); the GMCs were not statistically different for the interval groups (**Table G in S1 Data**).

In the 28- and 56-day groups of HIV-infected adults, patterns of responses were similar to those in healthy adults in terms of kinetics and magnitude (Fig 2B). Prior to dose 2, binding antibody responses were observed in 81% and 88% participants, with GMCs of 368 EU/mL (95% CI 272 to 497) and 291 EU/mL (95% CI 233 to 364) in the 28- and 56-day regimens, respectively (Table H in S1 Data). At 21 days post-dose 2, there was no evidence of a statistical difference in GMCs of binding antibodies (GMC ratio [95% CI] = 0.8 [0.6 to 1.1], p-value = 0.22) between the groups (Table G in <u>S1 Data</u>). The GMCs were 4,207 EU/mL (95% CI 3,233 to 5,474) and 5,283 EU/mL (95% CI 4,094 to 6,817) with 100% responder rates in the 28- and 56-day groups. At Day 365, antibodies persisted in 86% and 88% HIV-infected participants in the 28- and 56-day groups, respectively, with GMCs of 459 EU/mL (95% CI 352 to 600) and 338 EU/mL (95% CI 253 to 450) (Table H in S1 Data); no statistically significant difference (GMC ratio [95% CI] = 1.4 [0.9 to 2.0], p-value = 0.12) was observed between regimens. Compared to the healthy adults, the GMCs of binding antibodies at 21 days post-dose 2 were statistically different for the HIV-infected adults for both intervals (GMC ratio [95% CI] = 0.7 [0.5 to 0.99], p-value = 0.04 for 28-day interval; GMC ratio [95% CI] = 1.4 [1.1 to 1.9], *p*-value = 0.02 for 56-day interval; **Table I in** S1 Data).

**Response to booster dose.** In subsets of healthy participants from the 28- and 56-day interval groups, an Ad26.ZEBOV booster dose on Day 365 elicited anamnestic responses in both groups. GMCs rapidly increased, and, 7 days post-booster, levels were higher than after the initial Ad26.ZEBOV, MVA-BN-Filo regimen; 21 days post-booster, GMCs were 29,315 EU/mL (95% CI 20,614 to 41,689) in the 28-day group and 41,643 EU/mL (95% CI 32,045 to 54,116) in the 56-day group (**Table F in S1 Data**). In both groups, responses persisted in 97% to 100% of participants 1 year post-booster (Day 729) with similar GMCs (95% CI 4,383 to 4,534 EU/mL).

**Out-of-protocol window.** Among healthy participants, 204 (n = 169, active; n = 35, placebo) received their second dose outside the protocol-defined window. A sensitivity analysis of their binding antibody responses demonstrated that increases in the time interval between first and second doses from 99 to 483 days did not appear to negatively impact the immune response post-dose 2. After Ad26.ZEBOV vaccination, before MVA-BN-Filo vaccination, GMCs were maintained at similar levels for groups in which the interval was extended up to 140 days (266 EU/mL [95% CI 198 to 358]), 196 days (197 EU/mL [95% CI 159 to 245]), 252 days (239 EU/mL [95% CI 195 to 293]), or  $\geq$  280 days (212 EU/mL [95% CI 121 to 372]) (**Table J in S1 Data**). There was a trend for increasing responses 21 days after MVA-BN-Filo

vaccination as the interval lengthened. GMCs were 15,555 EU/mL (95% CI 10,907 to 22,184, n = 17) for 140 days, 14,995 EU/mL (95% CI 11,855 to 18,965, n = 52) for 196 days, 16,149 EU/mL (95% CI 13,882 to 18,786, n = 77) for 252 days, and 23,897 EU/mL (95% CI 16,703 to 34,188, n = 21) for intervals of  $\geq$ 280 days (**Table J in S1 Data**).

**Neutralising antibody responses against EBOV GP.** Neutralising antibodies against EBOV GP were tested in a subset of per-protocol analysis set healthy adult participants: 76 vaccinees and 21 placebo recipients (**Table K in S1 Data**). Neutralising antibodies were not observed in placebo recipients at any time point. At 21 days post-dose 2, 92% and 97% in the 28- and 56-day regimens, respectively, demonstrated neutralising antibodies against EBOV GP; geometric mean titres (GMTs) were 982 half maximal inhibitory concentration (IC<sub>50</sub>) (95% CI 714 to 1,350) and 4,100 IC<sub>50</sub> (95% CI 2,927 to 5,745) (**Fig 3 and Table K in S1 Data**). Titres decreased by Day 365 with 21% and 24% of the tested 28- and 56-day group participants, respectively, still displaying neutralising activity; GMTs were 123 IC<sub>50</sub> (95% CI <LLOQ–165) and 153 IC<sub>50</sub> (95% CI <LLOQ–210). There were strong correlations between the neutralising and binding antibody responses 21 days post-dose 2 (Spearman coefficient = 0.735 in 28-day interval group and 0.631 in 56-day interval group) (**S1 Fig**).

**Neutralising antibody responses against Ad26 vector.** Before vaccination, Ad26-specific neutralising antibodies were analysed in 98 healthy adults: 48 of the 28-day group (39 vaccinees,



Fig 3. Anti-EBOV GP neutralising antibody responses in healthy adults. The error bars represent the GMT ( $IC_{50}$ ) and its 95% CI. Subset of healthy adults administered with Ad26.ZEBOV on Day 1 and MVA-BN-Filo 28 or 56 days later as indicated. Responses in placebo groups are shown as open symbols. Black dotted line represents the LLOQ. Vaccines: Ad26 = Ad26.ZEBOV at a dose of  $5 \times 10^{10}$  vp; MVA = MVA-BN-Filo at a dose of  $1 \times 10^8$  Inf.U. CI, confidence interval; GMT, geometric mean titre;  $IC_{50}$ , half maximal inhibitory concentration; LLOQ, lower limit of quantification.

https://doi.org/10.1371/journal.pmed.1003813.g003

9 placebo) and 50 of the 56-day group (38 vaccinees and 12 placebo). An overall Ad26-specific seroprevalence rate of 91% with a GMT of 106  $IC_{90}$  (95% CI 79 to 142) was detected. Similarly, an overall 83% seroprevalence rate (GMT: 73  $IC_{90}$ ) was observed for the HIV-infected adults (**Table L in S1 Data**). Potential impact of Ad26-specific preexisting neutralising antibodies on vaccine-induced EBOV GP–specific humoral immune responses was examined at a participant level via descriptive statistical correlation analyses (**S2 and S3 Figs**). Negligible correlations were observed between Ad26-specific neutralising antibody titres at baseline and EBOV GP–specific binding antibody concentrations 21 days post-dose 2 (Spearman correlation coefficients: -0.02 and 0.22 for 28- and 56-day intervals in healthy participants, respectively; 0.02 and 0.01 for 28- and 56-day intervals in HIV-infected participants, respectively), as well as EBOV GP–specific neutralising antibody titres 21 days post-dose 2 (Spearman correlation coefficients: -0.15 and 0.11 for 28- and 56-day intervals, respectively).

#### Cellular immune responses against EBOV GP

Cellular immune responses were tested in a subset of healthy (16%) and HIV-infected (21%) participants. Twenty-one days post-dose 2, CD4+ T cell responses were observed in 32% (56-day interval) and 50% (28-day interval) of healthy participants. The median value tended to be numerically higher in participants vaccinated in the 28-day interval (0.11%) than the 56-day interval (0.06%) (S4A Fig). At the same time point, CD8+ T cell responses were detected in 29% (28-day interval) and 30% (56-day interval) of healthy participants (median value: 0.05%; < threshold, respectively) (S4B Fig).

In HIV-infected participants (56-day interval), 21 days post-dose 2, CD4+ T cell responses were observed in 40% of participants (median 0.18%). CD8+ T cell responses were observed in 17% of participants (median value <threshold) (S5A and S5B Fig). IFNγ ELISpot data (Table M in S1 Data) are described in S5 Text.

## Discussion

This Phase II study confirmed the safety, tolerability, and immunogenicity of heterologous 2-dose Ad26.ZEBOV, MVA-BN-Filo vaccination regimen against EBOV in African adults, including those with HIV infection. Safety and tolerability profiles were consistent with those observed in Phase I studies in European and African populations [15–17] and a Phase II study in a European population [22]. There were no deaths or SAEs associated with vaccination. Most AEs associated with either vaccine were transient and mild to moderate in severity, consisting mainly of injection-site pain, fatigue, headache, and myalgia.

As many conditions can influence the response to Ebola vaccines [23], it is important to ensure that local factors do not affect responses to vaccines. We previously reported the immune responses to the Ad26.ZEBOV, MVA-BN-Filo heterologous 2-dose regimen in Phase I studies in African populations [16,17]. The present study confirms these data and further demonstrates that HIV infection (well controlled by highly active antiretroviral therapy), a condition prevalent in African populations that could be expected to influence vaccine efficacy, did not have any apparent influence on those immune responses. The magnitude of the antibody responses 21 days post-dose 2 in our representative African population was similar to that observed in a Phase II study performed in Europe with a similar design, with comparable levels of EBOV GP–specific binding antibodies and virus neutralising activity 21 days post-dose 2 with 28- and 56-day vaccination intervals, measured in the same laboratory [22].

We have also reported safety and immunogenicity data of the 2-dose heterologous Ad26. ZEBOV and MVA-BN-Filo vaccination regimen, given 8 weeks apart, in a 2-stage trial in Sierra Leone. This study was initially foreseen as a safety and clinical efficacy study but was amended to become a safety and immunogenicity study after waning of the outbreak in Sierra Leone [24,25].

In addition, our study confirms the optimal interval between the 2 vaccinations, i.e., 56 days. Strong immune responses are observed with each of the 3 intervals tested, but increasing the interval from 28 to 56 days resulted in 2-fold higher levels of binding antibodies in healthy adults and 4-fold higher levels of neutralising antibodies. Prolonging the interval from 56 to 84 days did not further increase these immune responses, confirming the 56-day regimen as the optimal choice to achieve a high-response magnitude with a reasonably short vaccination schedule.

Some healthy adult participants either did not receive dose 2 or received dose 2 outside of their protocol-defined interval. However, sensitivity analyses of immune responses among participants who received their second vaccination "out-of-window" confirmed that extending the interval between Ad26.ZEBOV and MVA-BN-Filo vaccinations to >280 days resulted in immune responses that were at least as high in magnitude as the 56-day interval.

Although >90% of the adult participants had preexisting neutralising antibodies against the Ad26 vector, there was no indication that this had any impact on the vaccine-induced immune responses based on descriptive statistical analyses.

A mechanistic correlate of protection has not yet been determined, and there is no human correlate of protection against EBOV infection. As demonstrated in nonhuman primate lethal challenge models, EBOV GP–specific IgG binding antibodies correlated strongly with protection and, thus, can be considered as a measure of efficacy [26,27]. When measured from 10 days after a single dose of the rVSV-ZEBOV-GP vaccine in a ring vaccination strategy in the 2014 to 2016 and 2018 to 2020 outbreaks, protective efficacy was found to be 100% and 97.5%, respectively [7,8]. This efficacy was achieved with an rVSV-ZEBOV-GP vaccine formulation that has been reported to elicit a GMC of 1,262 EU/mL 28 days after vaccination in non-African populations [28]. This GMC was measured using the same assay in the same laboratory as our study, in which the Ad26.ZEBOV, MVA-BN-Filo vaccine regimen elicited EBOV GP–specific antibody concentrations from 3,085 to 7,518 EU/mL 21 days after the second vaccination in these African populations, depending on the interval between vaccinations.

One year post-Ad26.ZEBOV vaccination, binding antibodies were observed in 78% to 88% of participants. An Ad26.ZEBOV booster dose at 1 year elicited strong anamnestic responses with 55-fold increases in binding antibodies in both 28- and 56-day groups within 7 days of the booster dose. This observation suggests that the 2-dose Ad26.ZEBOV, MVA-BN-Filo vaccine regimen had established immune memory [29] that could rapidly be reactivated, which could be very important in the context of recurrent epidemics observed in Africa.

The numerous EBOV outbreaks in the DRC highlight the need for a prophylactic vaccine against this virulent disease. Current WHO SAGE recommendations are to use the single-dose rVSV-ZEBOV-GP vaccination for reactive use in those at high risk of contracting EBOV and prophylactic use and evaluation of the heterologous 2-dose Ad26.ZEBOV, MVA-BN-Filo regimen for those at less imminent risk [12]. Our study adds to the database of knowledge about this heterologous 2-dose strategy, particularly in the target population of African adults. The finding that increasing the interval between vaccinations from 56 to 84 days (and even further) resulted in similarly high binding antibody responses allows for potential flexibility in vaccinations, which might be useful in practice. The second part of our study in paediatric participants to ensure that the strategy is safe and immunologically effective in children as young as 4 years of age will be published separately. Additional studies are already underway to further evaluate the Ad26.ZEBOV, MVA-BN-Filo vaccination regimen in African populations, including younger children and infants, pregnant women, and healthcare workers who may have to travel to regions recurrently affected by outbreaks such as in the DRC.

Study limitations include that some healthy adult participants either did not receive dose 2 or received dose 2 outside of their protocol-defined interval. However, sensitivity analyses of immune responses among participants who received their second vaccination "out-of-window" showed that extending the interval between Ad26.ZEBOV and MVA-BN-Filo vaccinations to >280 days resulted in immune responses that were at least as high in magnitude as the 56-day interval. Another limitation is that the follow-up period was limited to 365 days for the majority of participants, and so it was not possible to determine whether immune responses persisted beyond this time period. However, this limitation could not be avoided as a follow-up period had to be determined before commencing the study, and, reassuringly, modelling results [30] suggest that immune responses persist beyond this time. Finally, no formal statistical testing of safety or immune response data was originally planned for this study. Although post hoc statistical comparisons were performed, no direct conclusions on the differences in safety and immunogenicity observed between regimens or between healthy and HIV-infected groups can be made; a clinically meaningful difference in terms of binding antibody levels is not known.

In conclusion, our study confirms that the heterologous 2-dose Ad26.ZEBOV, MVA-BN-Filo vaccination regimen has a safety, tolerability, and immunogenicity profile in African adults similar to that already demonstrated in European volunteers. These data supported the prophylactic indication against EBOV disease as licensed by the European Union.

## Supporting information

**S1 CONSORT Checklist.** (DOCX)

**S1 Study Protocol.** (PDF)

**S1 Text. Inclusion and exclusion criteria.** (DOCX)

**S2 Text. Participant subset selection criteria for exploratory assays.** (DOCX)

**S3 Text. Determination of neutralising antibody activity in an EBOV GP psVNA.** EBOV GP, Ebola virus glycoprotein; psVNA, pseudovirion neutralisation assay. (DOCX)

**S4** Text. Sample interpretation and responder definition for cellular immune assays. (DOCX)

**S5 Text. EBOV GP-specific IFNγ producing T cell responses (IFNγ ELISpot).** EBOV GP, Ebola virus glycoprotein; ELISpot, enzyme-linked immunospot; IFNγ, interferon gamma. (DOCX)

**S1 Data.** Table A. SAEs in the full analysis set after each vaccination dose, n (%). Table B. Solicited AEs\* in the full analysis set: Healthy adults stratified to 18–50 years and > 50 years age groups, n (%). Table C. Comparison (healthy adults versus HIV-infected adults) of solicited and unsolicited AEs in the full analysis set after each vaccination dose based on Fisher's exact test. Table D. Summary of fever in the full analysis set after each vaccination dose, n (%). Table E. EBOV GP–specific binding antibodies in healthy adult participants in the vaccine groups stratified to 18–50 years and >50 years age groups—per-protocol analysis set. Table F. EBOV GP– specific binding antibodies in healthy adult participants in the active vaccine groups—per-protocol analysis set. Table G. Comparison of EBOV GP–specific binding antibodies in healthy adult and HIV-infected adults in the active vaccine groups—per-protocol analysis set. Table H. EBOV GP-specific binding antibodies in HIV-infected adults in the active vaccine groups—per-protocol analysis set. Table I. Comparison of EBOV GP-specific binding antibodies in healthy adult versus HIV-infected adult participants in the active vaccine groups—per-protocol analysis set. Table J. EBOV GP-specific binding antibodies in healthy adults receiving the second dose of active vaccine outside the protocol-defined window. Table K. GMT of EBOV GP-specific pseudovirion neutralising antibodies titres (IC<sub>50</sub>) and responder rates in healthy adult active vaccine groups 1 and 2—per-protocol analysis set. Table L. Ad26 neutralising antibodies (Ad26 VNA, IC<sub>90</sub> titre) in healthy and HIV-infected adults—per-protocol analysis set. Table M. EBOV GP-specific IFN $\gamma$  producing T cell responses (IFN $\gamma$  ELISpot) in healthy and HIV-infected adults—per-protocol analysis set, Ad26 VNA, Ad26-specific virus neutralisation assay; AE, adverse event; EBOV GP, Ebola virus glycoprotein; ELISpot, enzyme-linked immunospot; GMT, geometric mean titre; IFN $\gamma$ , interferon gamma; SAE, serious adverse event. (DOCX)

**S1 Fig. Correlations between EBOV GP-specific binding and neutralising antibodies in participants assigned to an active regimen.** (A) 21 days post-dose 2. (B) 364 days post-dose 2. EBOV GP, Ebola virus glycoprotein; LLOQ, lower limit of quantification; ULOQ, upper limit of quantification.

(DOCX)

S2 Fig. Correlations between Ad26-specific neutralising antibody titres at baseline and EBOV GP-specific binding and neutralising antibodies 21 days post-dose 2 in healthy adults. (A) Anti-EBOV GP IgG ELISA at 21 days post-dose 2 by Ad26 neutralisation assay at baseline. (B) EBOV GP neutralisation assay at 21 days post-dose 2 by Ad26 neutralisation assay at baseline. Ad26 VNA, Ad26-specific virus neutralisation assay; EBOV GP, Ebola virus glycoprotein; ELISA, enzyme-linked immunosorbent assay; IgG, immunoglobulin G; LLOQ, lower limit of quantification; ULOQ, upper limit of quantification. (DOCX)

S3 Fig. Correlations between Ad26-specific neutralising antibody titres at baseline and EBOV GP-specific binding antibodies at 21 days post-dose 2 in HIV-infected adults by anti-EBOV GP IgG ELISA. Ad26 VNA, Ad26-specific virus neutralisation assay; EBOV GP, Ebola virus glycoprotein; ELISA, enzyme-linked immunosorbent assay; IgG, immunoglobulin G; LLOQ, lower limit of quantification; ULOQ, upper limit of quantification. (DOCX)

**S4 Fig. EBOV GP-specific CD4+ (A) and CD8+ (B) T cell cytokine responses in healthy adult participants.** EBOV GP, Ebola virus glycoprotein; n, number of participants with data at that time point; NA, not applicable. (DOCX)

**S5 Fig. EBOV GP-specific CD4+ (A) and CD8+ (B) T cell cytokine responses in HIVinfected adult participants.** EBOV GP, Ebola virus glycoprotein; n, number of participants with data at that time point; NA, not applicable. (DOCX)

## Acknowledgments

The authors wish to acknowledge the contribution of all participants who have taken part in the study; the partners of the EBOVAC2 consortium: the French National Institute for Health

and Medical Research (Inserm), the London School of Hygiene and Tropical Medicine, the University of Oxford, the Centre Muraz and Inserm transfert; the members of the Scientific Advisory Board; the members of Ethical Advisory Board; the members of the DSMB; the research staff and all members of the Clinical Operations Group at Janssen who worked on the trial (further information in supplement). The authors also thank Dr Michael Katwere for his facilitation of local clinical operations for the Sponsor. We thank Chelsea McLean (Janssen Vaccines and Prevention, Leiden, the Netherlands), Keith Veitch (Keithveitch Communications, Amsterdam, the Netherlands), Marialuisa Quadri (Janssen Vaccines and Prevention, Leiden, the Netherlands), Vikki Clayton (Ashfield MedComms, an Ashfield Health company) for medical writing assistance. We also thank Yvonne Salzgeber (Janssen Vaccines AG, Bern, Switzerland) and Sonia Silva (Janssen Vaccines and Prevention, Leiden, the Netherlands) for publication coordination.

The EBL2002 Study Group includes members of the EBOVAC 2 Executive Steering Committee (Pr Yves Levy [Inserm], Pr Peter Piot [London School of Hygiene and Tropical Medicine], and Pr Johan van Hoof [Janssen B.V.]); members of the EBOVAC 2 Clinical Steering Committee (Dr Macaya Douoguih [Janssen B.V.], Dr Cynthia Robinson [Janssen B.V.], Pr Rodolphe Thiebaut [Inserm], Dr Laura Richert [Inserm], Dr Deborah Watson-Jones [London School of Hygiene and Tropical Medicine], Pr Brian Greenwood [London School of Hygiene and Tropical Medicine], Pr Andrew J Pollard [University of Oxford], Dr Matthew D Snape [University of Oxford], Pr Nicolas Meda [Centre Muraz], Dr Houreratou Barry [Centre Muraz], Florence Chung [Inserm Transfert], and Sinéad Quigley [Inserm Transfert]); members of Janssen B.V. (Kim Offergeld [Clinical Program Leader, Global Clinical Operations], Benoit Callendret [Compound Development Team Leader], Stephanie Dincq [Clinical Project Management Lead], Camille Ferrault [Clinical Project Management Lead], Helga Pissens [Global Trial Leader], Marleen van Looveren [Regulatory Medical Writer], Sylvia van Ballaert [Regulatory Medical Writer], Tinne de Cnodder [Global Data Management Leader], and Tracy Hendrick [Clinical Programming Lead]); Clinical Working Group members (Sanne de Ridder [Clinical Trial Manager], Len Roza [ID&V Risk Management], Njinju Fogap [Lead Programmer], Rachana Gundluru [Senior Specialist Central Monitoring Manager], Olanrewaju Oladimeji [Global Data Manager], Vanessa Errijgers [Supply Management Coordinator], Maartje van Welij [Senior Independent Drug Monitor Manager], and Nicolette Muller [Clinical Research and Functional Manager]); the France EBL2002 coordination team (Anton Ottavi, PhD [EBOVAC2 Project Coordinator] and Eugénie Destandau [Communication Officer]); members of Inserm CIC 1401, EUCLID/F-CRIN Clinical Trials Platform (Christine Schwimmer, PhD [EUCLID Executive Director], Christine Betard [Clinical Trial Project Manager], Laetitia Moinot, PharmD [Clinical Trial Project Manager], and Cédrick Wallet [EUCLID Operations Manager]); members of INSERM U955, Vaccine Research Institute, Université Paris-Est Créteil (Aurélie Wiedemann [Immunologist] and Christine Lacabaratz [Immunologist]); Dr Albert Minga (Site Investigator) of Centre Médical de Suivi des Donneurs de Sang, Abidjan; Pr Désiré Blehou (Site Investigator) of Service de Santé au Travail SAPH, Toupah/Ousrou; members of PAC-CI, CHU Treichville, Abidjan (Dr Ida Viho, MD [Study Coordinator] and Dr Patrick Coffie [Expert]); Dr André Inwoley (Expert) of CeDReS, Université Félix Houphouët-Boigny, Abidjan; members of the KAVI Institute of Clinical Research, University of Nairobi (Nancy Thairu [Finance Manager], Dr Borna Nyaoke [Investigator], Dr Lavinia Bwisa [Investigator], Bashir Farah [Laboratory Manager], Moses Mundia [Data Manager], and Dorothy Essendi [Study Coordinator]); members of the Makerere University Walter Reed Project, Kampala (Dr Salim Wakabi [PI], Maureen Mukyala [Study Coordinator], Amir Wamala [PoR], Allan Tindikahwa [Regulatory and Compliance Manager], Dr Betty

Mwesigwa [Sub-Investigator], and Ezra Musingye [Data Manager]); members of the MRC/ UVRI and LSHTM Uganda Research Unit, Entebbe (Prof. Pontiano Kaleebu [Co-PI], Dr Jennifer Serwanga [Sub-Investigator/Immunologist], Dr Ggayi Abu-Baker Mustapher [Sub-Investigator/Project Coordinator], Dr Jonathan Kitonsa [Sub-Investigator/Study Clinician], Mr Paul Taire [Pharmacy Technician], Dr Laura Joan Nsangi [Sub-Investigator/ Study Clinician], Mr Vincent Basajja [Community Liaisons Officer], Mr Tobias Vudriko [Safety Laboratory Technologist], Mr Ben Gombe [Immunology Laboratory Technologist], Mr Hellen Kalungi [Study Nurse], Mr Francis Kasekende [Study Pharmacist], and Dr Mary Nyantaro [Sub-Investigator/Study Clinician]); members of Centre Muraz, Bobo-Dioulasso (Dr Lionel Wilfried Ouedraogo [Laboratory Manager], Dr Armel Poda [Sub-Investigator], Dr Bachirou Tinto [Pharmacist], Dr Dramane Kania [Laboratory Manager], Dr Naalona Sandrine Hien [Sub-Investigator], Dr Guekoun Lougue [Pharmacist], Dr Innocent Valea [Coordinator], Pr Halidou Tinto [Coordinator], and Dr Ines Evelyne Da [Sub-Investigator]); members of Groupe de Recherche Action en Santé (GRAS)/Centre National de Recherche et de Formation sur le Paludisme, Ouagadougou (Dr Alfred B. Tiono [Study Coordinator], Dr Alphonse Ouedraogo [Clinical Investigator], Dr Edith Bougouma [Study Pharmacist], Dr Issa Nebie [Senior Lab Manager], Dr Diarra Amidou [Lab Manager], and Dr Daouda Ouattara [Study Physician]); and the Independent Data Monitoring Committee (Dr Bruce Mcclain, Dr Geert Molendberghs, Dr Tsiri Agbenyega, and Dr Eric Wenceslas Joseph Balayssac).

## **Author Contributions**

- **Conceptualization:** Houreratou Barry, Gaudensia Mutua, Sodiomon B. Sirima, Laura Richert, Georgi Shukarev, Cynthia Robinson, Auguste Gaddah, Viki Bockstal, Kerstin Luhn, Rodolphe Thiébaut.
- **Data curation:** Hannah Kibuuka, Zacchaeus Anywaine, Sodiomon B. Sirima, Nicolas Meda, Omu Anzala, Serge Eholie, Christine Bétard, Laura Richert, Christine Lacabaratz, Georgi Shukarev, Cynthia Robinson, Auguste Gaddah, Dirk Heerwegh, Viki Bockstal, Kerstin Luhn, Maarten Leyssen, Rodolphe Thiébaut.
- Formal analysis: Kerstin Luhn.

Investigation: Macaya Douoguih.

Methodology: M. Juliana McElrath, Stephen De Rosa, Kristen W. Cohen.

Supervision: Kerstin Luhn, Rodolphe Thiébaut.

Visualization: Kerstin Luhn.

- Writing original draft: Houreratou Barry, Gaudensia Mutua, Hannah Kibuuka, Zacchaeus Anywaine, Sodiomon B. Sirima, Nicolas Meda, Omu Anzala, Serge Eholie, Christine Bétard, Laura Richert, Christine Lacabaratz, Georgi Shukarev, Cynthia Robinson, Auguste Gaddah, Dirk Heerwegh, Viki Bockstal, Kerstin Luhn, Maarten Leyssen, Macaya Douoguih, Rodolphe Thiébaut.
- Writing review & editing: Houreratou Barry, Gaudensia Mutua, Hannah Kibuuka, Zacchaeus Anywaine, Sodiomon B. Sirima, Nicolas Meda, Omu Anzala, Serge Eholie, Christine Bétard, Laura Richert, Christine Lacabaratz, M. Juliana McElrath, Stephen De Rosa, Kristen W. Cohen, Georgi Shukarev, Cynthia Robinson, Auguste Gaddah, Dirk Heerwegh, Viki Bockstal, Kerstin Luhn, Maarten Leyssen, Macaya Douoguih, Rodolphe Thiébaut.

#### References

- 1. Piot P, Spencer J. From 1976 to 2018: reflections on early investigations into the Ebola virus. Trans R Soc Trop Med Hyg. 2018; 112:527–8. https://doi.org/10.1093/trstmh/try088 PMID: 30137538
- Malvy D, McElroy AK, de Clerck H, Günther S, van Griensven J. Ebola virus disease. Lancet. 2019; 393:936–48. https://doi.org/10.1016/S0140-6736(18)33132-5 PMID: 30777297
- World Health Organization. Situation Report: Ebola Virus Disease 10 June 2016. Available from: https://apps.who.int/iris/bitstream/handle/10665/208883/ebolasitrep\_10Jun2016\_eng.pdf;jsessionid= D802F9B18350E8C97D0F9FA5FEEE6ABB?sequence=1.
- Ebola Outbreak Epidemiology Team. Outbreak of Ebola virus disease in the Democratic Republic of the Congo, April-May, 2018: an epidemiological study. Lancet. 2018; 392:213–21. https://doi.org/10.1016/ S0140-6736(18)31387-4 PMID: 30047375
- 5. World Health Organization. 10th Ebola outbreak in the Democratic Republic of the Congo declared over; vigilance against flare-ups and support for survivors must continue. [cited 2021 Sept 1]. Available from: https://www.who.int/news/item/25-06-2020-10th-ebola-outbreak-in-the-democratic-republic-of-the-congo-declared-over-vigilance-against-flare-ups-and-support-for-survivors-must-continue.
- Lévy Y, Lane C, Piot P, Beavogui AH, Kieh M, Leigh B, et al. Prevention of Ebola virus disease through vaccination: where we are in 2018. Lancet. 2018; 392:787–90. <u>https://doi.org/10.1016/S0140-6736(18)</u> 31710-0 PMID: 30104048
- Henao-Restrepo AM, Camacho A, Longini IM, Watson CH, Edmunds WJ, Egger M, et al. Efficacy and effectiveness of an rVSV-vectored vaccine in preventing Ebola virus disease: final results from the Guinea ring vaccination, open-label, cluster-randomised trial (Ebola Ça Suffit!). Lancet. 2017; 389:505– 18. https://doi.org/10.1016/S0140-6736(16)32621-6 PMID: 28017403
- 8. World Health Organization. Preliminary results on the efficacy of rVSV-ZEBOV-GP Ebola vaccine using the ring vaccination strategy in the control of an Ebola outbreak in the Democratic Republic of the Congo: an example of integration of research into epidemic response. [cited 2021 Mar 25]. Available from: https://www.who.int/csr/resources/publications/ebola/ebola-ring-vaccination-results-12-april-2019.pdf.
- FDA. First FDA-approved vaccine for the prevention of Ebola virus disease, marking a critical milestone in public health preparedness and response. Press Release 19 December 2019. [cited 2021 Mar 25]. Available from: https://www.fda.gov/news-events/press-announcements/first-fda-approved-vaccineprevention-ebola-virus-disease-marking-critical-milestone-public-health.
- Vaccine against Ebola: Commission grants first-ever market authorisation. Press Release 11 November 2019. [cited 2021 Mar 25]. Available from: https://ec.europa.eu/commission/presscorner/detail/en/ IP\_19\_6246.
- 11. World Health Organization. WHO prequalifies Ebola vaccine, paving the way for its use in high-risk countries. [cited 2021 Mar 25]. Available from: https://www.who.int/news-room/detail/12-11-2019-who-prequalifies-ebola-vaccine-paving-the-way-for-its-use-in-high-risk-countries.
- World Health Organization. Strategic Advisory Group of Experts (SAGE) on immunization interim recommendations on vaccination against Ebola Virus Disease (EVD) 7 May 2019. [cited 2021 Mar 25]. Available from: https://www.who.int/immunization/policy/position\_papers/interim\_ebola\_ recommendations\_may\_2019.pdf?ua=1.
- 13. Vaccine against Ebola: Commission grants new market authorizations. [cited 2021 Mar 25]. Available from: https://ec.europa.eu/commission/presscorner/detail/en/IP\_20\_1248.
- 14. "The Ministry of Health Launched UMURINZI Ebola Vaccine Program Campaign" Press release. [cited 2021 Mar 25]. Available from: http://rbc.gov.rw/index.php?id=100&tx\_news\_pi1%5Bnews%5D= 530&tx\_news\_pi1%5Bday%5D=14&tx\_news\_pi1%5Bmonth%5D=12&tx\_news\_pi1%5Byear%5D= 2019&cHash=9bcec711608382d977905e5f68b0e56a.
- Milligan ID, Gibani MM, Sewell R, Clutterbuck EA, Campbell D, Plested E, et al. Safety and immunogenicity of novel Adenovirus type 26- and Modified Vaccinia Ankara-vectored Ebola vaccines: a randomized clinical trial. JAMA. 2016; 315:1610–23. https://doi.org/10.1001/jama.2016.4218 PMID: 27092831
- Mutua G, Anzala O, Luhn K, Robinson C, Bockstal V, Anumendem D, et al. Safety and immunogenicity of a 2-dose heterologous vaccine regimen with Ad26.ZEBOV and MVA-BN-Filo Ebola vaccines: 12month data from a phase 1 randomized clinical trial in Nairobi, Kenya. J Infect Dis. 2019; 220:57–67. https://doi.org/10.1093/infdis/jiz071 PMID: 30796816
- Anywaine Z, Whitworth H, Kaleebu P, Praygod G, Shukarev G, Manno D, et al. Safety and immunogenicity of a 2-dose heterologous vaccination regimen with Ad26.ZEBOV and MVA-BN-Filo Ebola vaccines: 12-month data from a phase 1 randomized clinical trial in Uganda and Tanzania. J Infect Dis. 2019; 220:46–56. https://doi.org/10.1093/infdis/jiz070 PMID: 30796818
- DAIDS Adverse Event Grading Tables. [cited 2021 Sept 1]. Available from: <u>https://rsc.niaid.nih.gov/</u> sites/default/files/daidsgradingcorrectedv21.pdf.

- 19. US Food and Drug Administration. Guidance for industry: Toxicity grading scale for healthy adult and adolescent volunteers enrolled in preventive vaccine clinical trials. 2007 [cited 2021 Mar 25]. Available from: https://www.fda.gov/media/73679/download.
- Logue J, Tuznik K, Follmann D, Grandits G, Marchand J, Reilly C, et al. Use of the Filovirus Animal Non-Clinical Group (FANG) Ebola virus immuno-assay requires fewer study participants to power a study than the Alpha Diagnostic International assay. J Virol Methods. 2018; 255:84–90. <u>https://doi.org/ 10.1016/j.jviromet.2018.02.018</u> PMID: 29481881
- Horton H, Thomas E, Stucky JA, Frank I, Moodie Z, Huang Y, et al. Optimization and validation of an 8color Intracellular Cytokine Staining (ICS) assay to quantify antigen-specific T Cells induced by vaccination. J Immunol Methods. 2007; 323(1):39–54. https://doi.org/10.1016/j.jim.2007.03.002 PMID: 17451739
- 22. Pollard AJ, Launay O, Lelievre J-D, Lacabaratz C, Grande S, Goldstein N, et al. Safety and immunogenicity of a two-dose heterologous Ad26.ZEBOV and MVA-BN-Filo Ebola vaccine regimen: a phase 2 randomised clinical study in Europe (EBOVAC2). Lancet Infect Dis. 2020. doi: S1473-3099(20)30476-X
- 23. Gross L, Lhomme E, Pasin C, Richert L, Thiebaut R. Ebola vaccine development: systematic review of pre-clinical and clinical studies, and meta-analysis of determinants of antibody response variability after vaccination. Int J Infect Dis. 2018; 74:83–96. https://doi.org/10.1016/j.ijid.2018.06.022 PMID: 29981944
- Ishola D, Manno D, Afolabi MO, Keshinro B, Bockstal V, Rogers B, et al. Safety and long-term immunogenicity of a two-dose Ad26.ZEBOV, MVA-BN-Filo Ebola vaccine regimen in adults: a randomised, double-blind, controlled trial in Sierra Leone. Lancet Infect Dis. 2021. <u>https://doi.org/10.1016/S1473-3099(21)00125-0 PMID: 34529963</u>
- Afolabi MO, Ishola D, Manno D, Keshinro B, Bockstal V, Rogers B, et al. Safety and immunogenicity of a two-dose Ad26. ZEBOV, MVA-BN-Filo Ebola vaccine in children: a randomised, double-blind, controlled trial in Sierra Leone. Lancet Infect Dis. 2021. <u>https://doi.org/10.1016/S1473-3099(21)00128-6</u> PMID: 34529962
- 26. Callendret B, Vellinga J, Wunderlich K, Rodriguez A, Steigerwald R, Dirmeier U, et al. A prophylactic multivalent vaccine against different filovirus species is immunogenic and provides protection from lethal infections with Ebolavirus and Marburgvirus species in non-human primates. PLoS One. 2018; 13:e0192312. https://doi.org/10.1371/journal.pone.0192312 PMID: 29462200
- Marzi A, Engelmann F, Feldmann F, Haberthur K, Shupert WL, Brining D, et al. Antibodies are necessary for rVSV/ZEBOV-GP-mediated protection against lethal Ebola virus challenge in nonhuman primates. Proc Natl Acad Sci U S A. 2013; 110:1893–8. <u>https://doi.org/10.1073/pnas.1209591110</u> PMID: 23319647
- 28. Halperin SA, Das R, Onorato MT, Liu K, Martin J, Grant-Klein RJ, et al. Immunogenicity, lot consistency, and extended safety of rVSVΔG-ZEBOV-GP vaccine: a phase 3 randomized, double-blind, placebo-controlled study in healthy adults. J Infect Dis. 2019; 220:1127–35. https://doi.org/10.1093/infdis/jiz241 PMID: 31505665
- Balelli I, Pasin C, Prague M, Crauste F, Effelterre TV, Bockstal V, et al. A model for establishment, maintenance and reactivation of the immune response after vaccination against Ebola virus. J Theor Biol. 2020; 495:110254. https://doi.org/10.1016/j.jtbi.2020.110254 PMID: 32205143
- Pasin C, Balelli I, Van Effelterre T, Bockstal V, Solforosi L, Prague M, et al. Dynamics of the humoral immune response to a prime-boost Ebola vaccine: quantification and sources of variation. J Virol. 2019; 93:e0057. https://doi.org/10.1128/JVI.00579-19 PMID: 31243126